Author's response to reviews

Title: Health Beliefs and Carer Burden in First Episode Psychosis

Authors:

Maya Patel (map660@doctors.org.uk)
Rajan Chawla (rajan.chawla@nhs.net)
Carl R Krynicki (c.krynicki@bham.ac.uk)
Philip Rankin (PXR993@bham.ac.uk)
Rachel Upthegrove (r.upthegrove@bham.ac.uk)

Version: 4 Date: 4 June 2014

Author's response to reviews: see over
Dear Professor Korleila,

Re MS: 5719890321163248 - Health Beliefs and Carer Burden in First Episode Psychosis

Thank you for provisionally accepting our manuscript for publication and allowing this research to reach a wider audience.

In response to reviewer’s comments and the editorial requests, we have addressed these in turn. Changes are highlighted in the text.

Reviewer Suggestions:

We have addressed all minor revisions suggested and typographical errors.

Editorial Requests:

We have addressed all minor stylistic revisions.

The manuscript has been formatted entirely with BMC’s guidelines and all references formatted correctly. Many thanks again for accepting our manuscript for publication.

Yours Sincerely

Dr Rachel Upthegrove MBBS MRCPsych PhD